Author’s response to reviews

Title: The closed medial total subtalar joint dislocation without fracture of ankle case report

Authors:

AZARKANE Mohamed (azarkanemohamed02@gmail.com)
Boussakri Hassan (boussakri.med.1983@hotmail.com)
Alayyoubi Abdelghani (abdelghanielayoubi122@gmail.com)
Bachiri Mohamed (bachiri1985@yahoo.fr)
Elrhazi Amine (aelrhazi@yahoo.fr)
Abid Hatim (hatim.abid1@gmail.com)
Elidrissi Mohamed (elidrissitraumato@hotmail.fr)
Shimi Mohamed (shimimohamed@yahoo.fr)
Elibrahimi Abdelhalim (ehalim@gmail.com)
Elmrini Abdlemejid (abdlemjidmerini@yahoo.com)

Version: 3 Date: 8 March 2014

Author’s response to reviews: see over
Letter of motivation

Dear colleagues,

I wrote this article to present a rare emergency trauma case, which is the subtalar dislocation!

This is a rare disease often due to high energy trauma, which simply needs a simple reduction and cast immobilization under general anesthesia in often cases. But its prognosis remains limited to the cost of early reduction, sufficient immobilization to enable ligament healing and rehabilitation which should be early with sufficient number of sessions. This requires a shared responsibility between a competent doctor and a cooperating patient.

I chose the publication of this article in the journal case reports for reasons that this is a well known journal that takes interests in a rare case report and encourages scientific work.

With all my deepest respects

AZARKANE MOHAMED